Author's response to reviews

Title: Prospective study on the mismatch concept in acute stroke patients within the first 24h after symptom onset - 1000Plus study

Authors:

Benjamin Hotter (benjamin.hotter@charite.de)
Sandra Pittl (sandra.pittl@charite.de)
Martin Ebinger (martin.ebinger@charite.de)
Gabriele Oepen (gabriele.oepen@charite.de)
Kati Jegzentis (kati.jegzentis@charite.de)
Kohsuke Kudo (kokudo@iwate-med.ac.jp)
Michal Rozanski (michal.rozanski@charite.de)
Wolf U Schmidt (wolf.schmidt@charite.de)
Peter Brunecker (peter.brunecker@charite.de)
Chao Xu (chao.xu@charite.de)
Peter Martus (peter.martus@charite.de)
Matthias Endres (matthias.endres@charite.de)
Gerhard J Junghulsing (jan.junghulsing@charite.de)
Arno Villringer (arno.villringer@charite.de)
Jochen B Fiebach (jochen.fiebach@charite.de)

Version: 3 Date: 12 November 2009

Author's response to reviews: see over
Dear Editorial Production Team,

We applied the following formatting changes according to your revisions:

Major revisions:
- Figure 1 was cited at the end of chapter Methods/Design – Patients.
- The tables were added to the manuscript and should not be added as additional files but be integral part of the publication. Please excuse the confusion.
- Table 1 was already cited, Table 2 is now referred to at the end of chapter Discussion – Selection.

Minor revisions:
- The manuscript title has been changed.
- Methods heading has been changed.
- Typography has been checked for errors.
- Additional files: the EndNote library will be deleted in the course of submission.

We hope to have met your demands for the manuscript – if not, please don’t hesitate for any further enquiries or comments.

Kind Regards,

Dr. Benjamin Hotter
CSB, Resident

Berlin, November 12th 2009
Covering letter for our revised manuscript 1615190615286955